Author's response to reviews

Title: Empathy in senior year and first year medical students: A cross-sectional study

Authors:

Eunice Magalhães (eunicemagalhaes@ecsaude.uminho.pt)
Ana P Salgueira (anasalgueira@ecsaude.uminho.pt)
Patrício Costa (pcosta@fpce.up.pt)
Manuel J Costa (mmcosta@ecsaude.uminho.pt)

Version: 4 Date: 19 July 2011

Author's response to reviews:

Dear Sirs,

Please find on this site the manuscript with revisions (R3) “Empathy in senior year and first year medical students: A cross-sectional study” by Eunice Magalhães et al and a point-by-point description of the changes that were made by authors. We are hopeful that format revisions will be sufficient and all typographical errors were corrected.

Point-by-point description of the changes:

• Page 4: references were corrected and cited in numerical order. Reference 6 was replaced by 5 and the subsequent references were ordered numerically.

• Page 5; line 16: the repeated word "are" was removed.

• Page 10, line 7: the misspelling “qith” was replaced by “with”.

• Page 15, line 3: the misspelling “resluts” was replaced by “results”.

• Page 16; line 3: the repeated word "of" was removed.

Thank you very much,
I look forward to hear from you.
Sincerely,
Eunice Magalhães